Author's response to reviews

Title: The relation between insulin resistance and lung function: a cross sectional study

Authors:

Gül Sagun (gulsagun@yahoo.com)
Canan Gedik (drcanangedik@gmail.com)
Esra Ekiz (dr.esra@gmail.com)
Engin Karagoz (dr.engindhly@yahoo.com)
Mumtaz Takir (mumtaztakir@yahoo.com)
Aytekin Oğuz (aytekinoguz@hotmail.com)

Version: 5 Date: 7 September 2015

Author's response to reviews:

STROBE guideline was filled.
Authors contribution, Competing interest, Acknowledgement sections were divided.